Author's response to reviews

Title: Analysis of the contribution of FTO, NPC1, ENPP1, NEGR1, GNPDA2 and MC4R genes to obesity in Mexican children: a case-control study

Authors:

Aurora Mejía-Benítez (amejia@cinvestav.mx)
Miguel Klünder-Klünder (klunderk@gmail.com)
Loic Yengo (loic@good.ibl.fr)
David Meyre (meyred@mcmaster.ca)
Celia Aradillas (celia@uaslp.mx)
Esperanza Cruz (mcruzl@yahoo.com)
Elva Pérez-Luque (elvaleticiaperez@yahoo.com)
Juan M Malacara (jmmalacara@hotmail.com)
Maria E Garay (enia911@hotmail.com)
Jesús Peralta-Romero (driperalta@hotmail.com)
Samuel Flores-Huerta (floreshuertamd@gmail.com)
Jaime García-Mena (jgmena@cinvestav.mx)
Philippe Froguel (p.froguel@imperial.ac.uk)
Miguel Cruz (mcruzl@yahoo.com)
Amélie Bonnefond (amelie.bonnefond@good.ibl.fr)

Version: 2 Date: 28 July 2012

Author's response to reviews:

Hi

Following an email from the editorial office (that you can find below), we have added the ethics statement in the methods of the present manuscript.

Best regards
Amélie Bonnefond (corresponding author)

MS: 2042165653771946

Research article

Analysis of the contribution of FTO, NPC1, ENPP1, NEGR1, GNPDA2 and MC4R genes to obesity in Mexican children: a case-control study

Aurora Mejía-Benítez, Miguel Klünder-Klünder, Loic Yengo, David Meyre, Celia Aradillas, Esperanza Cruz, Elva Pérez-Luque, Juan M Malacara, Maria E Garay, Jesús Peralta-Romero, Samuel Flores-Huerta, Jaime García-Mena, Philippe Froguel, Miguel Cruz and Amélie Bonnefond

BMC Medical Genetics

Dear Dr. Bonnefond,
Thank you for your recent submission to BMC Medical Genetics. Before we can proceed with peer review, can you please include an ethics statement into your manuscript. We strongly urge you to make this promptly, as we cannot start the peer review process until we have received a version containing the statement. Please refer to our standard policy below:

“Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.”

Once you have clarified this, please upload the revised version by following these instructions:

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted.
2. Click on the 'Submit revision' button next to the title of this manuscript.
3. In the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version.
4. In the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.
5. In the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments.

We would be grateful if you could resubmit your files with the changes requested within the week. If we haven't heard from you or received the updated version by that time, I'm afraid we may need to close the file, as we cannot keep submissions on hold indefinitely.

Please do not hesitate to contact us if you have any questions.

Best wishes,

*Guian Paolo Declaro*

Journal Editorial Office

*Bio**Med****Central*

Web: biomedcentral.com